Author's response to reviews

Title: Comparing exercise interventions to increase persistence with physical exercise and sporting activity among people with hypertension or high normal blood pressure: protocol for a randomised trial

Authors:

Chris Fife-Schaw (c.fife-schaw@surrey.ac.uk)
Simon de Lusignan (s.lusignan@surrey.ac.uk)
Joe Wainwright (j.wainwright@surrey.ac.uk)
Hannah Sprake (h.sprake@surrey.ac.uk)
Suzannah Laver (suzannah.laver@surreycc.gov.uk)
Victoria Heald (victoria.heald@surreycc.gov.uk)
Julian Orton (julian.orton@gp-h81091.nhs.uk)
Matt Prescott (matt@isostasy.co.uk)
Helen Carr (hfcarr@doctors.org.uk)

Version: 2
Date: 17 January 2014

Author's response to reviews:

Hi. There is a typo in the article title - it should not have the word 'cluster' in it.